Author's response to reviews

Title: Assessment of occupational injuries among Addis Ababa city municipal solid waste collectors: A cross-sectional study

Authors:

Daniel Bogale (bogaleodo@gmail.com)
Abera Kumie (aberakumie2@yahoo.com)
Worku Tefera (workutefera2000@yahoo.com)

Version: 2 Date: 22 August 2013

Author's response to reviews:

Figures and tables are separated (tables are placed at the end of the reference as it was before and figures are uploaded separately as two figures).

English version questionnaire and observation checklist are uploaded as an additional files separately.

On the authors detail academic status of one of the author is corrected (previously it was written as if he was PhD holder but he is PhD fellow).